Contents lists available at ScienceDirect



Environmental Innovation and Societal Transitions

journal homepage: www.elsevier.com/locate/eist

Policy mixes across vertical levels of governance in the EU: The case of the sustainable energy transition in Latvia





Inese Zepa^{*}, Volker H. Hoffmann

Department of Management, Technology, and Economics, ETH Zurich, Weinbergstrasse 56/58, Zurich 8092, Switzerland

| ARTICLE INFO | A B S T R A C T |
|--|---|
| <i>Keywords:</i> Sustainability transitions Energy policy Vertical policy mixes Latvia | The European Green Deal requires policy alignment horizontally across policy areas but also vertically across governance levels. Accordingly, the diversity of actors and their preferences can hinder sociotechnical transitions. We, therefore, ask: How do policy mixes unfold across vertical scales of governance in sociotechnical transitions? We analyse vertical policy mix characteristics, considering consistency, coherence, comprehensiveness, and credibility, and the inherent feed-back loops between politics and policy. Drawing on empirical evidence from Latvia, gathered through content analysis and interviews with senior stakeholders, we highlight two key findings: First, friction points in policy coherence between the national-local governance levels feed back into the vertical policy mix, as EU-level policy strategies are not substantiated with corresponding policy instruments. Second, the lack of political commitment to the transition at the national level perpetuates friction points in comprehensiveness and credibility of the vertical policy mix, hindering transition processes. |

1. Introduction

Climate change and environmental degradation remain pressing global challenges, also to be addressed in the EU. In response, the European Green Deal aims to accelerate decarbonisation to attain climate neutrality by 2050, particularly focusing on the energy sector (European Commission, 2022b). Catalysing technological innovation, especially technology diffusion for the energy transition, in turn, requires policy alignment across policy areas and levels of governance. However, this alignment is currently lacking, for instance, in advancing renewable energy technologies (Markard and Rosenbloom, 2020).

The sustainability transitions literature has explored policy alignment using the concept of 'policy mixes', referring to the combination of policy instruments and processes through which such instruments emerge and interact (Rogge and Reichardt, 2016). Research has focused on policy mixes spanning various sectors or technologies at one level of governance, most often the state (Capano and Howlett, 2020; Kivimaa and Kern, 2016; Kivimaa and Virkamäki, 2014; Ossenbrink et al., 2019).

However, policy mixes span various governance levels, creating complexity and friction between them. Sustainability transitions research has largely neglected this verticality aspect of policy mixes (Flanagan et al., 2011; Howlett et al., 2017). Political scientists have evoked the concept of 'multi-level governance' to consider various relationships across levels of governance. However, the approach does not account for the complexity of policy mix dynamics in sociotechnical transitions that emerges through feedback mechanisms between politics and policy across governance levels (Edmondson et al., 2019). We aim to fill the gap and unpack vertical

* Corresponding author.

E-mail address: inese.zepa@alumni.ethz.ch (I. Zepa).

https://doi.org/10.1016/j.eist.2023.100699

Received 17 August 2022; Received in revised form 25 January 2023; Accepted 29 January 2023

Available online 10 February 2023

^{2210-4224/© 2023} The Authors. Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

policy mixes (VPM) spanning supranational, state, and local governance levels for sustainability transitions. Specifically, we ask: How do policy mixes unfold across vertical scales of governance in sociotechnical transitions?

Our analysis is structured by the logic of policy mix characteristics, drawing on the seminal work on policy mixes by Rogge and Reichardt (2016) and Edmondson et al. (2019). For policy mix characteristics, we distinguish the 'consistency' of policy elements (policy strategies and instruments), the 'coherence¹' of policy processes (policy making and implementation), and the 'comprehensiveness', and 'credibility' of the overall policy mix. We also identify feedback mechanisms between politics and policy across supranational, state, and local governance levels.

We conduct an in-depth analysis of the electricity sector transition in Latvia. Latvia is part of the EU and is mandated to accelerate the sustainable energy transition in line with the European Green Deal (European Commission 2020). Increasing the share of renewables requires diversification into new technologies like wind and solar PV, which are currently negligible in Latvia's energy mix. However, the supporting policies for such technologies in Latvia have been minor, endangering the achievement of the EU's Green Deal ambition. Essentially, friction points emerge in sustainability transition processes across governance levels. The findings of this paper can be extended to other situations where a country that is part of a supranational entity is lagging behind in a particular policy area.

We rely on multiple qualitative methods: semi-structured interviews with senior policymakers and energy sector stakeholders at the three governance levels as well as content analysis. The analysis focuses on the timeline starting with the EU Renewable Energy Directive in 2018 (2018/2001/EU) that sets the first binding renewable energy target for 2030 (later revised upwards in the 'Fit for 55' package) (European Parliament, 2021).

We primarily aim to make a theoretical contribution to sustainability transition policy and politics research, offering a new perspective on operationalising policy mix characteristics across governance levels. We also make an empirical contribution, exploring vertical policy mixes in a novel geographical context for sustainability transition scholarship. We highlight two findings: First, friction points in policy coherence between the national-local governance levels feed back into the vertical policy mix, as EU-level policy strategies are not substantiated with corresponding policy instruments. Second, the lack of political commitment to the transition at the national level perpetuates friction points in comprehensiveness and credibility of the vertical policy mix, hindering transition processes.

The paper is structured as follows: Section 2 provides a brief literature review and outlines the theoretical framework. Section 3 elaborates on the methodology, and Section 4 gives a background to the case. Section 5 presents the findings, followed by a discussion in Section 6. Section 7 concludes the paper.

2. Theoretical framework

2.1. Vertical policy mixes

Governance plays a crucial role in context-dependant transition processes, given the high uncertainty over future technological, economic, and political developments (Kern et al., 2019; Kuzemko et al., 2016). Single policies are insufficient to achieve the complex sustainability transition goals. The sustainability transitions literature has increasingly engaged with policy mixes, defined as "interactions and interdependencies between different policies as they affect the extent to which intended policy outcomes are achieved" (Flanagan et al., 2011, p. 702). For a comprehensive view of policy mixes, we distinguish 'policy elements' and 'policy processes'. Policy elements include policy strategies and specific policy instruments, which are "concrete tools to achieve overarching objectives" (Rogge and Reichardt, 2016, p. 1623). Policy processes, including policy planning and implementation, refer to the "political problem-solving process among constrained social actors in the search for solutions to societal problems" (Rogge and Reichardt, 2016, p. 1625).

Policy mix research has largely focused on policy elements, although policy processes also form an indispensable part of policy mixes. "Behind policy there is always politics" (Meadowcroft, 2011, p. 73), as policy elements are shaped by policy processes (Howlett and Rayner, 2007; Markard et al., 2012). In other words, policy mixes are rarely free from political challenges caused by diverse preferences of actors with varying degree of power (Flanagan et al., 2011; Rogge et al., 2017). As a result, policy mixes often evolve haphazardly, resulting in incompatible policy layering or insufficient policies to destabilise the incumbent industries (Howlett et al., 2017; Kivimaa and Kern, 2016).

Scholars have explored 'horizontal' policy mixes² across policy areas (Del Río, 2014; Flanagan et al., 2011; Grubb et al., 2017; Magro and Wilson, 2019). Example works have assessed policy coherence between emissions trading and renewable electricity support schemes (Del Río, 2007), energy, climate, and security policy (Kivimaa and Sivonen, 2021; Strambo et al., 2015), and R&D policies (Nauwelaers et al., 2009), among others.

We also identified research in 'vertical' policy mixes, primarily in federal country contexts (Huang, 2019; Kuzemko et al., 2020; Magro and Wilson, 2013). Lanahan and Feldman (2015) and Huang (2019) both find that bottom-up as well as top-down factors shape policy mixes across levels of governance in federal countries. Lanahan and Feldman (2015) particularly highlight how state government policy responses complement federal level policies, drawing on the example of the US Small Business Innovation Research

¹ Note that scholars sometimes use the word 'coherence' referring to policy outputs (similar to the way 'consistency' is used in this paper), e.g., Kivimaa and Sivonen (2021); Nilsson et al. (2012)

² Early work explores interactions between policies, without explicitly referring to 'policy mixes'

| | | | | ▲ Feedback loops |
|------------|----------------|---------------------|------------------|--|
| | | | SN Supra NA | anational governance level National governance level LO Local governance level |
| Policy | elements | | Policy processes | |
| Strateg | gies | Instruments | Planning | Implementation |
| | | <u>SN, NA, LO</u> 1 | SN-NA, NA-LO | |
| I Co | nsistency | | II Coherence | |
| SN-NA-LO 2 | | | | ' |
| | mprehensivenes | S | | |
| IV Cre | edibility | | | |

Fig. 1. Policy mix characteristics, adapted from Rogge & Reichardt (2016).



Fig. 2. Operationalising policy mixes, extended from Rogge & Reichardt (2016), Edmondson et al. (2019).

programme. Huang (2019) shows how national priorities can reshape policy mixes at lower governance levels. For instance, urban governments have been more active in mobilising resources than provincial governments for solar water heating technology development in China. However, the existing analyses have not engaged with the supranational governance level in vertical policy mixes. Sustainability transitions research tends to submerge that as a 'landscape' factor. Unpacking the whole verticality of policy mixes can render helpful insights.

Scholars from a different field, political science, have studied 'multi-level governance' to analyse relationships across governance levels, especially in the EU. These approaches explore how state governments develop and implement policies considering the supranational influences in international environmental regimes. For instance, scholars have explored how various domestic factors, such as the quality of democracy, affect member-state compliance with EU policies and laws (Andonova, 2004; Chaisty and Whitefield, 2015; Schimmelfennig and Sedelmeier, 2005). Scholars have also explored the making of supranational strategies in the first place (Ćetković and Buzogány, 2019; Knill and Liefferink, 2021). More recent literature has studied 'transnational' governance, paying closer attention to various actor types in contexts of diffused authority (Andonova and Tuta, 2014; Hale, 2020). Scholars have, for instance, analysed how smaller governance units, such as cities, can counterbalance inaction at higher governance levels (Borgström, 2019; Dale et al., 2020; Ehnert et al., 2018).

However, the approach does not do justice to the complex interaction mechanisms between policy elements and policy processes in

sociotechnical transitions across all governance levels (Cherp et al., 2011; Lindberg et al., 2019; Loorbach, 2010). Various friction points and feedback loops can distort a vertical policy mix, such as discrepancies in transition strategies and instruments across governance levels. Complex policymaking faces the risk of failure when vertical dimensions are not well integrated (Edmondson et al., 2020; Howlett et al., 2017; Howlett and Del Río, 2015). In a nutshell, political science scholars have engaged with verticality across governance levels without explicitly considering policy mixes for particular transition goals (e.g., energy). Sustainability transitions scholars have engaged with policy mixes for particular goals but not yet the full spectrum of governance levels. This paper aims to bridge this gap.

2.2. Operationalising vertical policy mixes

We aim to extend the current work to consider the policy mix dynamics across all governance levels: supranational, national, and local. We conceptualise the vertical policy mix, combining work on policy mix characteristics by Rogge and Reichardt (2016) and feedback mechanisms by Edmondson et al. (2019). The analysis relies on insights from assessing policy mix 'characteristics' (Fig. 1) – 'consistency' of policy elements (I), 'coherence' of policy processes (II), 'comprehensiveness' (III), and 'credibility' (IV) of the policy mix and the inherent feedback loops (1&2) between these across governance levels.

This paper offers an analytical framework for operationalising and assessing the policy mix characteristics across vertical governance levels (Fig. 2). We define assessment criteria for each policy mix characteristic (on a scale from low to high). The policy mix characteristics descriptions are extrapolated from a variety of sources. First, we transformed the definitions of characteristics used in Rogge and Reichardt (2016) into more actionable criteria. Second, we matched these criteria with views from the policy analysis literature. Finally, we triangulated our view on the criteria in discussions with colleagues at several academic conferences.

Regarding policy mix characteristics, consistency of policy elements (I) refers to how policy elements work together to achieve the policy objectives (Lindberg, 2019). We consider consistency in policy strategies (policy objectives with quantified targets and principal plans for achieving them), in policy instruments (as defined before), and between policy strategies and instruments. Policy element consistency requires little or no contradictions in strategic priorities (a), substantiated with fitting (b) and mutually reinforcing (c) policy instruments (considering their goals, type, and design features) (Bach and Hansen, 2021; Rogge and Reichardt, 2016).

Coherence of policy processes (II) refers to policy making, implying "processes that determine the elements of the policy mix" (Rogge and Reichardt, 2016, p. 1625) and policy implementation, namely, executing and enforcing the policy instruments. Policy processes are coherent when policy making and implementation are systematic and synergetic to achieve the policy objectives. The coherence should hold across policy planning (d), policy coordination (e), and also communication (f) (Ashoff, 2005; OECD, 2019b; Tanaka et al., 2020).

The comprehensiveness of the policy mix (III) refers to how extensive and exhaustive the policy elements and policy processes are. Comprehensive policy mixes address market, institutional, and systemic policy failures (g), considering technology-push and demandpull aspects (h) and leveraging economic, regulatory, and soft instruments (i) (Bach and Hansen, 2021; Costantini et al., 2017; Sovacool, 2009).

The credibility of the policy mix (IV) is also assessed across the entire vertical policy mix, considering policy elements (strategies and instruments with their design features) and processes (Rogge and Dütschke, 2018). Credible policy mixes signal reliability of policy commitment (j) to the transition, enabling a policy framework that is stable but also sufficiently flexible for any changes required (Newell and Goldsmith, 2001; Reichardt et al., 2016; Rogge and Reichardt, 2016).

Moreover, policy elements and processes are interlinked through feedback mechanisms. Friction points in the policy mix can trigger negative feedback loops, as policies are not only products of politics but also influence subsequent politics (Kingdon and Stano, 2011; Wright, 2008). Essentially, "policy mixes stimulate changes in sociotechnical systems through policy effects, and [...] these changes can subsequently generate feedback mechanisms influencing the evolution of the policy mix" (Edmondson et al., 2019, p. 2).

For the purposes of this analysis we distinguish two feedback loops. First, consistent policy elements facilitate coherent policy processes that, in turn, reinforce further consistency in policy elements (1 in Figs. 1 and 2) in case of a positive feedback loop. In case of negative feedback loop, low policy element consistency undermines policy process coherence and thus further policy element consistency. In other words, policies facilitating the transition empower niche stakeholders (e.g. through material, ideational, and institutional effects) increasing their political participation that, in turn, contributes to more systematic policies for sustainability transitions (Bolwig et al., 2019; Edmondson et al., 2019; Oberlander and Weaver, 2015). Second, multiple such positive feedback loops between policy elements and processes can contribute to comprehensiveness and credibility of the overall policy mix (e.g. pro-transition actors securing stronger socio-political, fiscal, administrative influence). A more comprehensive and credible policy mix, in turn, reinforces policy process coherence and element consistency (2 in Figs. 1 and 2). In case of a negative feedback loop, lacking political commitment to the transition lowers comprehensiveness and credibility of the vertical policy mix, hindering change. In other words, politics shapes policy, shapes politics.

This paper explores the phenomenon in the context of vertical policy mixes, as policy elements and processes must align across the supranational, national, and local governance levels. Although the state remains the primary energy policy decision-maker, friction points in policy mix consistency and coherence can emerge vertically between supranational and national (friction point highly relevant for policy making) and national and local governance levels (friction point highly relevant for policy implementation), triggering negative feedback loops for the overall comprehensiveness and credibility of the policy mix. Note that the focus of this paper is to assess vertical policy mixes across governance levels, but policy mixes have to be consistent, coherent, comprehensive, and credible at each governance level. This paper speaks to the horizontal policy mixes only where that has direct implications for the vertical policy mix.

List of interviewees.

| Code | VPM level | Organisation | Role |
|------|---|--|--|
| EU1 | EU | DG Energy | Former Senior Official |
| EU2 | EU | European Commission | Economic Advisor |
| EU3 | EU | DG Energy | Country Expert |
| N1 | State | Ministry of Economics | Director of Department |
| N2 | State | Ministry of Economics | Director of Department |
| N3 | State | Ministry of Economics | Head of Unit |
| N4 | State | Ministry of Economics | Head of Unit |
| N5 | State | Ministry of Environmental Protection and Regional Development | Adviser to the Minister |
| N6 | State | Bank of Latvia | Senior Official |
| N7 | State | Electric utility company | Head of Division |
| L1 | Local | Riga City Council | Councillor |
| L2 | Local | Riga Energy Agency | Senior Official |
| L3 | Local | EU-driven innovation initiative | Managing Director |
| 01 | Other | Stockholm School of Economics | Head of Department |
| 02 | Other | International energy company | Head of Division |
| | Code EU1 EU2 EU3 N1 N2 N3 N4 N5 N6 N7 L1 L2 L3 O1 O2 | CodeVPM levelEU1EUEU2EUEU3EUN1StateN2StateN3StateN4StateN5StateN6StateN7StateL1LocalL2LocalL3LocalO1OtherO2Other | CodeVPM levelOrganisationEU1EUDG EnergyEU2EUEuropean CommissionEU3EUDG EnergyN1StateMinistry of EconomicsN2StateMinistry of EconomicsN3StateMinistry of EconomicsN4StateMinistry of EconomicsN5StateMinistry of EconomicsN6StateBank of LatviaN7StateElectric utility companyL1LocalRiga Energy AgencyL3LocalEU-driven innovation initiativeO1OtherStockholm School of Economics |

3. Methodology

3.1. Case selection

We adopt a single case study design to analyse the vertical policy mix across three governance levels. Such a single case study approach enables in-depth exploratory, context-specific analysis (George and Bennett, 2005; Yin, 2009).

We draw on empirical evidence from Latvia's electricity sector transition. The energy sector in Latvia necessitates a fundamental energy sector transition towards renewable energy technologies (Ministry of Economics LV 2020). The electricity sector is a fundamental pillar in the NECP, and the greening of other sectors such as transport relies on the supply of clean and affordable electricity. On the one hand, the supranational EU level strongly influences energy sector policy at state level, also unlocking various support mechanisms, including financial (Solorio Sandoval and Jörgens, 2017). On the other hand, support policies for niche technologies in Latvia, particularly wind and solar PV have been minor (as outlined in the National Energy and Climate Plan, NECP) (EBRD 2016). With that, the case of electricity transition in Latvia can be instrumental for also understanding other cases where state and local level policy support for EU level ambition is missing, threatening to derail, here, the European Green Deal.

3.2. Data collection

We rely on multiple qualitative methods to collect data for the analysis of the vertical policy mix: content analysis and semistructured interviews.

As a first step to outlining the relevant policy elements (policy strategies and instruments) and policy processes (facts on policy planning and implementation), we conducted a content analysis (Newbold et al., 2002; Stemler, 2000). To identify the policy elements, we analysed and evaluated 25 policy documents (considering documents relevant to the current policy mix, starting with the Renewable Energy Directive published in 2009), including policy strategies and plans across all three governance levels (list of sources in the Appendix). In addition, we conducted an extensive press search on the electricity sector transition in Latvia to identify the policy processes and gain additional insight on policy elements. We used the Factiva search tool to identify news articles in English between 2018 and 2021. Relying on the selected criteria, we identified 673 articles on the topic and read 312 articles relevant to the electricity sector transition in Latvia (excluding false positives and articles with repetitive information, criteria are outlined in the Appendix). Moreover, we also read news articles from three widely read news sources in Latvia (in the Latvian language, covering the same period): weekly news periodical 'ir', online news 'delfi.lv', and Latvian TV and radio portal 'lsm.lv'.

In the second step, we conducted 15 semi-structured interviews with senior stakeholders at all three governance levels. (Table 1)³ lists the interviewees, their organisations and roles. We selected the interviewees based on a systematic scanning of profiles to include both stakeholders that take part in policy processes (e.g., government officials) and stakeholders that benefit (or do not) from the policy elements (e.g., energy producers). The highest number of interviews was with stakeholders at state level, which remains the main governance level in energy policy processes. We also incorporated perspectives from stakeholders at supranational and local levels (the latter focusing on Riga municipality, the largest local administrative unit that offers most data points). We relied on snowball sampling where appropriate and interviewed an independent expert to discuss our first findings. Given the country's small size, the count of interviews was sufficient to cover all necessary angles. The interviews were conducted in 2021 using online conferencing tools. With one exception, we conducted all interviews in the native language (Latvian), which helped build rapport with

³ Interview identities are presented at a high level of abstraction to protect their confidentiality



Fig. 3. Electricity production in Latvia, ktoe (Eurostat, 2020b).

the interviewees, increasing clarity and precision. Each interview lasted 45–60 min, and we recorded the interviews upon interviewees' consent.⁴ We transcribed all the interviews. Altogether we compiled and worked with 97 pages of notes and transcripts. The interview guidelines and coding frames reflect the analytical framework outlined in Section 2. We adjusted the interview guidelines to fit each interviewe's expertise and reflect prior findings.

3.3. Data analysis

Building upon the insights from content analysis and interviews with senior stakeholders, we looked for evidence on the vertical policy mix assessment criteria introduced in Fig. 2. We assessed the characteristics on a low to high scale, based on the descriptions of the end-points. For instance, for consistency of policy elements (I), we searched for evidence on the alignment of strategic priorities (a), instruments that serve these strategies (b), and alignment among instruments (c). We assessed consistency as rather high if we identified alignment in these criteria across governance levels (or low if we found evidence that policy strategy and elements undermine each other). We repeated the process for all policy mix characteristics across vertical governance levels (illustrative interview guidelines in the Appendix).

In the last step, we triangulated our findings by reviewing reports by international organisations relevant to Latvia's energy sector transition. Latvia is not a member of the International Energy Agency (IEA), so comprehensive external energy sector analyses are not available. However, other organisations, such as the Organization for Economic Co-operation and Development (OECD), offer analyses of various aspects relevant to the electricity sector transition in the country.

4. Case description

Having regained independence from the Soviet Union in 1990, Latvia has undergone significant political and economic changes in the past three decades, including in the energy sector. Nevertheless, the historic energy sector integration with the Soviet Union has shaped energy sector dynamics, for instance, perpetuating the dependence on Russian gas (Grigas, 2013b).

Although the share of renewable energy and particularly electricity in Latvia's energy mix is above the EU average, increasing that share any further (in line with the European Green Deal) necessitates structural changes in the energy sector. Renewable sources constitute 53% (vs 38% EU average) of gross final electricity consumption in 2020 (most of the remaining electricity is generated in two combined heat and power plants (CHPs), powered by natural gas) (Eurostat, 2020a). Latvia's target is to reach over 60% of gross final electricity consumption in 2030 (Ministry of Economics LV 2020). However, nearly 40% of electricity in 2020 is generated in three large hydropower plants on the Daugava river (Eurostat, 2020b). Any further increase in the share of renewables requires diversifying into cost-efficient electricity generating technologies, primarily wind and solar energy. Currently, wind and solar technologies constitute a negligible share of Latvia's energy mix (2% in 2020, growing only at 1% p.a. since 2015, see Fig. 3) (Eurostat, 2020b). Accordingly, energy transition requires a significant change from the status quo. The geographic conditions for the technologies are favourable in Latvia, as confirmed by feasibility studies and installed capacities in neighbouring countries, such as

⁴ Privacy and confidentiality questions were covered at the start of each interview, all interview notes and recordings are kept on secure university servers

Existing and planned policies in Latvia's electricity transitions¹¹ (Ministry of Economics LV 2020).

| Main policy instruments ^a | Brief description | Timeline |
|--|--|-----------------------------|
| Mandatory procurement mechanism | Compensation mechanism for producers of electricity from renewable sources and in cogeneration plants | Existing ^b |
| Implement transnational projects for offshore wind parks | One high capacity wind park (at least 800 MW), co-financed by Connecting Europe Facility and Common Interest Projects (3.1) | Planned (2030) |
| Review territorial construction regulation and land use restrictions for RES technologies | Revise rules (incl. clarifying general construction regulations, issuing building permits), map available to project developers regarding suitable areas (3.2) | Planned (2022) |
| Develop a conceptual solution for the development of land wind parks | Concept and legislative changes for large capacity wind parks (at least 800MW incl. the offshore wind park), additional 1600 GWh of wind energy per year (3.3) | Planned (2022) |
| Promote the use of solar electricity generation | Assess possibility for a loan/guarantee programme, e.g., from EU Structural Funds (3.4) | Planned/ approved (2022) |
| Carry out necessary assessments for further development of RES electricity | Assess possibility of technologically neutral RES auctions for newly installed capacity, evaluate possible incentive measures for municipalities (3.5) | Planned (2023) |
| Promote marketing of RES electricity | Develop framework for including electricity source information in invoices, RES electricity purchase contracts, trading of RES electricity (3.6) | Planned (2022) |
| Support development of innovative solutions to increase share of RES in energy system | Develop appropriate legislation, leverage EU Structural Funds (3.7) | Planned (2030) |
| Develop legal framework for promoting self- generation and self-consumption of energy | Develop appropriate legislation (4.1) | Planned (2023) |
| Promote development of energy communities | Develop appropriate legislation, carry out a study on viable business model (4.3) | Planned (2026) |
| Review the rates of the natural resource tax and the conditions for their application | Assess and develop appropriate legislation (11.2) | Planned (2025) |
| Improve public knowledge, awareness of RES use | Information campaigns, incl. on the role and necessity of the use of RES (12.1) | Planned (2023) |
| Legislation for system of guarantees of origin of electricity | Develop legislation to improve the system of guarantees of origin of electricity, integrate the system in single European System of Contributions of Energy (H6) | Planned (2022) |
| Accelerate procedures for RES technology deployment | Single point of contact for authorisation of RES deployment and publicly available instructions for the required authorisation, framework to enable local communities to benefit from RES (H7) | Planned (2022) |
| Funding to promote use of RES | Assess the possibility of various initiatives e.g. RES Promotion and Energy Efficiency Improvement Fund, selling RES statistics and channelling revenues to the Fund (H8) | Planned (2022) |

^a Planned policy measures according to NECP Appendix IV

^b Note: by the time of publication, the mandatory procurement mechanism might be abolished

Lithuania (Eurostat, 2021; Lindroos et al., 2018).

Latvia's energy sector is embedded in the EU energy sector strategies, also regarding the transition to sustainable energy sources in the framework of the European Green Deal. The EU single electricity market has contributed to energy diversification in Latvia through facilitating critical infrastructure projects (e.g. electricity connections to Estonia funded by Connecting Europe Facility European Commission, 2021a). In light of EU policies, Latvia has also liberalised the electricity market and Latvia's electricity grids will be synchronised with the EU network by 2025 (away from the grids of the Commonwealth of Independent States) (AST, 2022; Morgan, 2018).

Furthermore, the revised Renewable Energy Directive and 'Fit for 55' package sets a binding renewable energy target for the EU. As a member state, Latvia also partakes in decision-making processes at EU level through the Council of the EU, the European Parliament, as well as the European Commission (EC). Member states must reflect a sufficient level of ambition in their strategic planning line with this target, considering status quo domestic energy market realities (European Commission 2019). At the time of writing, most planned policy activities in Latvia as outlined in the NECP are pending further assessment and development (summarized in Table 2).

5. Findings

Our findings show the shortcomings in the vertical policy mix for the electricity sector transition in Latvia. The findings are structured by vertical policy mix characteristics, assessing the consistency of policy elements (5.1), coherence of policy processes (5.2), and comprehensiveness (5.3) and credibility (5.4) of the overall policy mix. Each sub-section starts with an overview table with the findings and illustrative quotes from the interviews, including supporting evidence from the Factiva search where helpful. Then, we discuss each assessment criteria separately and include references to the relevant feedback loops

5.1. Consistency of policy elements

We find that contradictions in policy strategies and instruments (and between the two) reduce policy element consistency in the vertical policy mix for Latvia's electricity sector transition (Table 3). The strategic ambition for sustainability transition at the supranational level is only partly reflected in strategies at the state and local levels. Moreover, the strategic ambitions are not substantiated with policy instruments and these at times undermine each other. Such contradictions limit resource (e.g., funding) and institutional (e.g., regulatory) support for niche technologies and relevant actors (with implication for Feedback loop 1 in Fig. 2).

First, alignment of strategic priorities is lacking across governance levels in Latvia's vertical policy mix for electricity transition. In principle, the goals defined in Latvia's NECP for 2030 (50% share of renewable energy and 60% share of renewable electricity in final

Consistency of policy elements.

| (I) Consistency of p | (I) Consistency of policy elements | | | | | | |
|---|------------------------------------|---|--|--|--|--|--|
| Assessment criteria | VPM friction point | Findings | Illustrative interview quotes and evidence | | | | |
| Alignment of strategic priorities | SN—NA | The higher ambition level of strategies at supranational level is insufficiently reflected in strategies at state level | "One can create structures for fundamental change, or pick low hanging fruit – Latvia does the latter []" (EU1) Latvia has largely implemented the European Commission recommendations in the energy sector, although on renewable energy most policies might not be sufficient to achieve the 2030 target (LETA, 2021) | | | | |
| | NA-LO | Only the largest municipalities have developed specific strategies | "Most municipalities do not have strategic energy policy plans or instruments in place []" (N5) | | | | |
| Instrument- strategy alignment | SN—NA | Supranational (and state level) strategic ambition is not sufficiently substantiated with policy instruments | "Interesting situation, where we run [renewable energy] projects in Estonia and Lithuania, but not in Latvia, because there is no support for it" (N7) Out of 800 new contracts for sale of solar panels and parts of solar parks in 2020 by Latvenergo, ¾ of total capacity is installed for customers outside Latvia (LETA, 2021) Latvia lacks long-term strategy for energy and instruments to achieve the plans, according to the Head of Latvian Wind Energy Association (LETA, 2021) | | | | |
| | NA-LO | Existing policy instruments contradict the supranational (and state level) strategic ambition ^a | "The mandatory procurement mechanism was neither a logical nor a fair policy instrument" (L1) The Audit Office released a report on Latvia's renewable energy policy, outlining unjustified support to cogeneration plants, incl. payment of EUR 500'000 to two companies in 2018/19 (LETA, 2021) 27 MPC licenses revoked for 'green' power plants after inspection (LETA, 2019) Government subsidies suspended for 21 green energy producers in 2019, saving EUR 31.5mn (LETA, 2020) | | | | |
| Inter-instrument alignment | SN—NA | Policy instruments curated at supranational level lack reinforcement at state level | "According to the EU, the [money from] the EU Cohesion Fund, should be complemented with other investment. Our reality is such that [] what the EU does not finance, that is left for the state and is usually not happening" (N4) | | | | |
| | NA-LO | Policy instruments at local level contradict policy instruments at state level | "Municipalities have the right, especially in protected areas, to trigger extra environment protection mechanisms (e.g., biodiversity screening) [] Territorial planning policies at municipality level is a key barrier [to wind energy projects]" (N1) | | | | |

^a Here consider lack of consistency emanating from policy design (e.g. policy instrument access critera), rather than from policy processes (e.g. implementation)

consumption) satisfy the EU criteria⁵ and reach above the EU average in absolute terms (European Commission 2020). Certain goals in the NECP also reverberate across other state-level strategies, e.g. energy sector innovation is also one of five 'smart specialisation' focus areas in Latvia's industrial strategy (in line with the EU Cohesion policy) (Cross-Sectoral Coordination Center 2020; Ministry of Education and Science LV 2018). However, the state level strategy fails to deliver concrete objectives (e.g., quantified trajectories) and the NECP is "picking low hanging fruit" (EU2), e.g., for solar and wind energy development. Moreover, only the largest municipality of the capital city Riga has a substantial local level strategy (Riga City Sustainable Energy and Climate Plan 2030) outlining objectives of reaching climate neutrality by 2050 (Ekodoma 2021).

Second, the supranational and national strategic ambition is not sufficiently substantiated with supporting policy instruments at national and local levels, limiting support for niche technologies and relevant actors. According to the EC assessment of Latvia's NECP, Latvia could miss its 2030 energy transition target without outlining targeted policy measures (European Commission 2020; OECD, 2019a). Stakeholders confirm that specific policy instruments are expected in the updated NECP version, but such have already been introduced in the neighbouring countries, for example, Lithuania (EU2). Moreover, the main policy instrument in Latvia's electricity sector transition, the mandatory procurement component (MPC) has conflicted with the strategic transition goals in practice (SPRK 2016). The MPC instruments speaks to the policy strategy "on paper, not in spirit" (L1). The MPC was designed to support renewable electricity producers but also (efficient) natural CHPs. Although the EC has approved the MPC instrument in theory, in practice (often inefficient) CHPs received the highest support share, an outcome driven by policy instrument design (e.g. access criteria) (Dreblow et al., 2013; Ministry of Economics LV 2018). As a result, wind and solar niche technologies and respective stakeholders didn't meaningfully benefit from this (financially costly) policy (Ministry of Economics LV 2020).

Third, missing inter-instrument alignment further limits support for niche technologies and actors. For instance, Latvia gains

⁵ In line with the formula in Annex II to Regulation 2018/1999 on the Governance of Energy Union and Climate Action, the ambition level was increased after the EU's first comments on the NECP draft

Coherence of policy processes.

| (II) Coherence of polic Assessment criteria | y processes VPM friction point | Findings | Illustrative interview quotes and evidence |
|--|---|---|---|
| Alignment of strategic planning | SN—NA | Effective alignment of strategic planning between supranational and state governance levels | "EC controls whether Latvia lives up to the minimal requirements, and in general, these strategic processes are well aligned" (EU2) |
| | NA-LO | Various political preferences and vested interests hinder policy planning (and implementation) at state and local governance levels | "Latvia has a significant fossil energy lobby that is counteracting renewable energy projects as much as possible. One cannot deny or ignore this. Gas is in a stable position with a large systemic influence. One could trace how the old energy sources are fighting the new." (EU2) |
| Alignment of coordination | SN—NA | Effective policy coordination at supranational and state governance levels | "[Policy coordination] has always been good, in NECP discussion meetings, in Brussels" (EU3) |
| | NA-LO | Policy coordination at the state and local governance levels is fragmented | "Currently many small uncoordinated initiatives, that is not logical [] At national level, it is not clear what the goal is [] Cannot accidentally arrive at the required outcome" (N6) Economics Ministry has prepared amendments to the electricity market law, proposing to scrap MPC (LETA, 2019) |
| Alignment of communication | SN—NA | Policy communication at the supranational-state nexus reinforces policy implementation | 85% of Latvia's residents support solar power development in Latvia, 77% wind energy (LETA, 2019) |
| | NA-LO | Policy communication complicates policy implementation at state and local governance levels | "For 10–15 years, we have not gotten rid of the green oligarch image in Latvia; when people think of a wind generator, they think of oligarchs, although that is no longer the case" (L1) "There are many myths about Latvia as a green country that data-based facts and indicators disprove" (EU2) |

significant funding for sustainability transition⁶ from the EU Cohesion Fund (CF) and the Recovery and Resilience Facility (RRF).⁷ However, funding for specific projects from the EU is often not complemented with funding options at the state level, reinforced by the generally low propensity to borrow: "Latvia has careful politics [...] we try to get by with what we have [...] also the capital market is shallow" (N6). Furthermore, institutional barriers at local level can contradict policy instruments at state level. Specifically, territorial planning in municipalities, often shaped by NIMBY effect, affects wind project development (Aboltiņš, 2019; Ministry of Economics LV 2020). For instance, a wind project in Tukums municipality was restricted at a late stage of project development, in a way that was later deemed illegal (N2, EU2). Although some initiatives at the EU level address such barriers, additional policy instruments at state and local levels are required to practically support niche technologies: "we need various mechanisms: upgrade legislation, create benefits for communities [...], involve inhabitants" (L1).

5.2. Coherence of policy processes

We identify friction points in policy process coherence, especially between national-local governance levels (Table 4). We find that vested interests distort strategic policy planning, policy coordination is fragmented, and policy communication does not reinforce public support for renewable energy technologies. Such policy processes reduce the bottom-up political commitment to the transition (with implication for Feedback loop 1 in Fig. 2).

First, strategic planning activities are not fully aligned across governance levels in Latvia's electricity policy mix, as status quo actors block systematic and synergetic policy support for the transitions. On the one hand, interviewers confirm that strategic planning activities between the EU and the national level function well (Carbon Market Watch 2018; European Parliament, 2021). On the other hand, at the state level the current government is the first to prioritise the energy transition in practice (EU1). Vested business interests have hindered strategic planning towards sustainability at the state and local levels, the topic has been researched by scholars and investigative journalists (Grigas, 2013a; Sprance, 2020). Specifically, there has been a strong gas lobby of Latvijas Gāze (natural gas company, partially owned by Gazprom⁸) in energy policy making, perpetuating support for the gas sector and electricity generation in CHP power plants (L1). For instance, Latvia was the last country in the EU to liberalise the gas market (Brauna et al., 2022). Accordingly, the policy processes are rather reactive to EU directives rather than strategically integrating the energy sector transition with broader economic development objectives (N1). It wasn't until 2021 that Latvenergo pledged to shift to renewable energy in Latvia (Irbe, 2022).

Second, policy coordination suffers from lack of involvement of the local governance level. Stakeholders confirm effective policy

⁶ According to the NECP, the transition in Latvia is expected to cost \sim 3% GDP per year (EUR 8 billion for the period 2021-30) (Kamenders et al., 2020)

⁷ 50% of CF funding in Latvia (EUR 468 million) and 42% of RRF (EUR 281 million) is allocated to climate and (primarily) energy efficiency projects (N4)

⁸ Latvia has relied on gas from Russia, wherey Gazprom held an exclusive supply contract until 2017 (Åboltinš & Akule, 2014)

Comprehensiveness of policy mix.

| · · · · · · · · · · · · · · · · · · · | 5 | | |
|--|---|--|--|
| (III) Comprehensiveness of th Assessment criteria | ne policy mix VPM friction point | Findings | Illustrative quotes and evidence |
| Market, institutional, and systemic failures addressed | SN—NA-LO | Market, institutional, and systemic failures hindering innovation in the energy sector not comprehensively addressed | "We are in the last place in the EU regarding innovation investment for state-owned companies [] There are limitations in the existing legislation" (N7) |
| Technology-push and market-pull effects addressed | SN—NA-LO | Lagging support measures to encourage market- pull for new technologies | "Need to support not suppliers but consumers" (EU1) One third of Latvia's households would be ready to generate their own electric power, majority through solar panels (according to Enefit survey) (LETA, 2020) 61% of Latvia's households changed electricity supplier in 2018 [accordingly, could be incentivized to think more about energy sources] (LETA, 2019) |
| Economic, regulatory, and soft instruments addressed | SN—NA-LO | The potential of economic instruments not fulfilled | "There are talks about an additional tax for renewable energy that would disincentivise investors" (O2) |

coordination at the supranational-state governance nexus (e.g., incorporating EC recommendations on the NECP draft European Commission, 2021b) (EU2, EU3). However, "municipalities are detached from energy policy making" (N5), leaving important actors that could favour transition policies out of the debate. To illustrate, the coordination with municipalities is the responsibility of the Ministry of Environmental Protection and Regional Development (VARAM), while energy and innovation policy making is the responsibility of the Ministry of Economics (EM) (L1). Stakeholders repeatedly outlined that "there is no dialogue between EM and VARAM" (L1). Although the National Energy and Climate Council, led by the Prime Minister, has been established to make coordinated policy proposals and advance innovation and financing, many stakeholders caution of the marginal benefit of the institution for niche stakeholders (Cabinet of Ministers LV 2019) (N1, N5, N6, EU1).⁹

Finally, policy communication also remains weak due to the low involvement of pro-transition voices. Despite the pro-transition communication from the EU level, e.g. through the EC representation in Latvia communicating the European Green Deal to businesses and the general public, it is essential to engage consumers regarding greener and potentially cheaper electricity options. Nevertheless, effective policy communication at state and local levels is lacking (L1). Additionally, the MPC was communicated as green energy support while de facto supporting natural gas, rendering a negative public image of renewable energy (as intransparent, excessively expensive) (Åboltiņš, 2019; Sprance and Kļava, 2019). "In Latvia, MPC is like swearing" (O2), fuelling consumer distrust in the energy sector.

5.3. Policy mix comprehensiveness

We find that the overall vertical policy mix is not sufficiently comprehensive to satisfy the strategic sustainability transition objectives (Table 5). First, market, institutional, and systematic sustainability transition aspects have not been comprehensively addressed, such as supporting the innovation of new technologies in Latvia's energy sector. Second, the market-pull elements require more support in Latvia's energy sociotechnical system, including targeted support for energy communities. Third, the potential of different kinds of policy instruments, including economic, regulatory, and soft instruments, is not exhausted, such as taxation policy. Essentially, lacking political commitment to the transition fails to ensure comprehensive policy mix (that, in turn, reinforces the low political commitment, with implications for Feedback loop 2 in Fig. 2).

First, market, institutional, and systemic failures hindering innovation in the energy sector are not comprehensively addressed. Namely, although strategically innovation in the energy sector is a policy priority, there is no comprehensive political and financial support for energy innovation at the state and local levels. The EC assessment of Latvia's NECP highlights the missing policy instruments to advance technological innovation in the energy sector,¹⁰ these are in the planning stage at the time of writing (N3) (European Commission 2020). One of the main barriers is that state-owned companies cannot purchase innovative technologies without a complex procurement procedure, which makes the process unviable (N7, N3, EU2). Accordingly, Latvia remains "in the last place in the EU regarding innovation investment for state-owned companies" (in contrast to the state-owned energy company Ignitis in Lithuania or examples from Nordic countries) (N7).

Second, market-pull measures specifically remain weak in Latvia's energy system. Lacking support for energy communities is a key example. According to the revised Renewable Energy Directive (2018/2001/EU), member states should ensure appropriate support schemes for energy communities and various projects at the EU level contribute to that (e.g., Energy Communities Repository

⁹ Note: by the time of publication, Latvia has created a new Ministry for Climate and Energy to combine the energy-related questions previously with EM and climate-related questions previously with VARAM (Kincis, 2023). However, the contribution of this institution remains to be seen. ¹⁰ More detailed measures are expected in the 2021-27 national industry policy guidelines

¹¹ Note: by the time of publication, some advances on the policies planned for 2022 have been made, partly accelerated by the pressures on the energy sector following the Russian invasion of Ukraine. These updates do not change the findings of this paper.

| Credibility of polic | y mix. | | |
|---|--|---|---|
| (IV) Credibility of t Assessment criteria | the policy mix VPM friction point | Findings | Illustrative interview quotes and evidence |
| Reliability of commitment | SN—NA-LO | Lacking political commitment at state and local levels decreases the reliability of the vertical policy mix | "It has always surprised me why [energy transition] comes so late. Energy is expensive, there is energy dependence on Russia. It would be an obvious choice [] it's been 30 years since independence" (O1) "Laws are changed, there are many legal processes, tainted legal framework, which is important for new local and foreign investment [] energy market situation is unattractive to foreign investors [] The gordian knot has not been broken. The past influences the present" (EU2) |

providing technical and administrative advice) (European Commission, 2022a). The NECP outlines the objective to increase the self-consumer share and update the enabling framework for energy communities. However, more targeted policies are pending at the state and local levels, especially given the high upfront cost of technologies (L1, EU1, N1).

Finally, the potential of diverse policy instruments is not realized, including economic instruments. A prime example is tax policy. The revised EU Energy Tax Directive defines the rules for electricity and energy products. Namely, tax breaks for fossil fuels should be phased out by 2030 (Latvian Wind Energy Association, 2022). To date, however, the tax system in Latvia provides indirect support for gas and CHPs: "without crippling the market, whereby gas is mysteriously cheaper, renewable energy would have gone up" (L1). Also, the tax is disproportionately high for low-income households (OECD, 2019a). Stakeholders remain sceptical about the usefulness of currently planned taxation in advancing renewable energy technologies (L1).

5.4. Policy mix credibility

Finally, with lacking political commitment to the transition at the state level, the policy mix also lacks credibility (Table 6). Policy elements developed at supranational level can reinforce the reliability of commitment to the transition at state and local governance levels. However, lacking socio-political, fiscal, and administrative feedbacks to steadily support transition policy at the state and local levels counteract the positive effects from the EU policies (Feedback loop 2 in Fig. 2).

On the one hand, the European Green Deal offers the required "push from abroad" (O1) for policy change to enable the sustainable electricity sector transition in Latvia. The EU strategies set clear expectations and timelines and provide financial and other types of policy support. On the other hand, lacking coherence in policy decision-making and vested interests lower the political commitment to the transition at state and local levels. In the context of frequently changing coalition governments and fragmented energy sector governance, predictability and continuity in energy sector policies remain low (O1, EU2). That, in turn, reduces the reliability of policy support for niche stakeholders. The frequent changes to the MPC policy instrument eligibility criteria, funding strategies, and ultimately phasing out trajectories illustrate this point. Overall, there has been no clear state politics: "political will is very important, we lack that" (L1).

6. Discussion

We analysed the consistency of policy mix elements, coherence of policy processes, as well as the comprehensiveness and credibility of the vertical policy mix for the electricity sector transition towards sustainability in Latvia. The policy mix characteristics shape feedback mechanisms between policy and politics, defining the vertical policy mix. In this section, we discuss our central findings, analyse the inherent feedback mechanisms, and deduce policy implications.

First, friction points in policy coherence between the national-local governance levels feed back into the vertical policy mix, as EUlevel policy strategies are not substantiated with corresponding policy instruments. We find discrepancies in policy strategies across governance levels, as the ambition for transition processes in supranational strategies does not resonate in strategic objectives and policy instruments that bring them to life at state (and local) governance levels. The relatively low policy element consistency in the vertical policy mix fails to create a supportive environment for niche technologies and with that, restricts the development of niche stakeholders. Accordingly, they do not gain the weight to contribute to policy process coherence at the state governance level. In the case of Latvia, the NECP is picking the 'low hanging fruit' without clear policy objectives and instruments to advance technological innovation to ensure the transition to sustainable electricity in line with the 2030 and ultimately, 2050 net zero goals.

Second, the lack of political commitment to the transition at the national level perpetuates friction points in comprehensiveness and credibility of the vertical policy mix, hindering transition processes. Vested interests, particularly the gas lobby, hinder systematic and synergetic policy making, especially at the state-local governance nexus. Within the broader political context of frequently changing coalition governments in the country, the overall comprehensiveness and credibility of vertical policy mix remains low even with strong EU commitment to the transition. That, in turn, perpetuates low policy process coherence and policy element consistency, triggering a negative feedback loop that results in unfavourable conditions for niche technologies. To illustrate, investors prefer to invest in renewable technology parks in neighbouring Lithuania, where the political conditions are better. Accordingly, the share of

wind and solar PV technologies has grown slowly in Latvia (Ministry of Economics, 2020b). This dynamic further slows the sociotechnical change towards the European Green Deal ambition.

Our findings on the feedback loops in the vertical policy mix for Latvia's electricity transition raise the question of what can be done to help generate political momentum in member-states for the required changes towards the European Green Deal. Even more so, given that comparable changes have been made in neighbouring member-states, for instance, Lithuania (Zepa, 2022). We identify two main policy implications. First, initiatives at supranational level can create incentives for political commitment to the transition at state and local levels. We observed that financial support from the EU is frequently a defining factor for energy transition initiatives. Further non-financial initiatives at EU level can also be helpful, such as targeted educational campaigns to shift consumers to green energy sources (e.g., informing on opportunities for energy communities). That, in turn, gives a stronger voice to pro-transition actors domestically, increasing political commitment to the transition. Second, a higher diversity of stakeholders, including from municipalities and regions, require substantial incentives and opportunities to partake in strategic policy planning at the state level. Increasing stakeholder participation in policy planning and following up on implementation can help unlock comprehensive and credible vertical policy mixes.

Our research also indicates that external shocks can affect these feedback loops. Especially the sharp increase in gas prices resulting from the Russian invasion of Ukraine in February 2022 has shaped energy policy in Europe with two-fold effects (Oxford Institute for Energy Studies 2022). On the one hand, shocks can perpetuate negative feedback loops by increasing support for non-sustainable alternatives, for instance, re-opening coal power plants in Germany (Connoly, 2022). On the other hand, external shocks can intensify seeking viable solutions for energy security, independence, and sustainability (IEA, 2022). For instance, Latvia has banned imports of Russian gas as of January 1st 2023 (Klūga, 2022). However, it remains to be seen whether the momentum can increase policy process coherence and ultimately, political commitment, shifting the feedback loops more in favour of sustainable energy transition.

Moreover, this paper renders avenues for future research. First, we made an early attempt to operationalise policy mix characteristics systematically. Further work would be helpful to scrutinise and detail the operationalisation of policy mix characteristics to devise a consistent approach helpful for empirical studies. This conceptual approach also requires broader empirical testing in comparative case studies across geographies and technologies. Second, we engaged with feedback loops at a high level of abstraction, identifying two mechanisms. Further work is required to conceptualise these in more detail across governance levels, also considering policy sequencing aspects (Meckling et al., 2017). Finally, more systematic approach to understanding vertical policy mix implications from external shocks would be insightful. Specifically, offering policy advice on reducing negative fluctuations in a (vertical) policy mix.

7. Conclusions

How do policy mixes unfold across vertical scales of governance in sociotechnical transitions? We explored the policy mix characteristics and the inherent feedback loops, drawing on data and insights from a comprehensive single case study analysis in the EU member state Latvia. We relied on semi-structured interviews and content analysis to gain detailed insights into the vertical policy mix.

Our analysis renders two key insights on policy mixes across vertical scales. First, friction points in policy coherence between the national-local governance levels feed back into the vertical policy mix, as EU-level policy strategies are not substantiated with corresponding policy instruments. Second, the lack of political commitment to the transition at the national level perpetuates friction points in comprehensiveness and credibility of the vertical policy mix, hindering transition processes. Essentially, the frequently changing policy strategies and instruments at state and local levels offset the ambitious sustainability agenda at EU level, reinforcing the status quo over substantial political and economic change.

Our findings suggest policy implications for overcoming friction points in the vertical policy mix and directions for future research. These are important to address for the EU Green Deal ambition.

Declaration of Competing Interest

We have no conflict of interest to disclose.

Data availability

Data will be made available on request.

Appendix

Appendix para (Tables A1-A5).

Table A1

Policy elements: policy documents analysed.

| Author EU governance level | Nr | Policy documer | nt | | |
|--|----------|------------------|---|--|---|
| European | 1 | Communication | Communication from the Commission: The European Green Deal (COM/2019/640) | | |
| Commission | 2 | A Framework S | A Framework Strategy for a Resilient Energy Union with a Forward-Looking Climate Change Policy (COM/2015/080) | | |
| | 3 | Renewable Ene | Renewable Energy Directive (2009/28/EC) | | |
| | 4 | Renewable Ene | rgy Dir | ective (2018/2021/EU, Clean Energy | y for all Europeans package) |
| | 5 | Baltic Sea Offsh | ore Wi | nd Joint Declaration of Intent | · · · · · · · · · · · · · · · · · · · |
| | 6 | Governance of | the Ene | rgy Union and Climate Action (EU) | Regulation 2018/1999 |
| | 7 | Publication: Cle | ean Ene | rgy for all Europeans | • |
| | 8 | Assessment of t | he Fina | l National Energy and Climate Plan | of Latvia |
| | 9 | Country Report | Latvia | (2020 European Semester) | |
| Other | 10 | BEMIP Offshore | e Wind | Work-program | |
| | 11 | Roadmap on Re | egional | Gas Market Integration between Es | onia, Finland, Latvia, and Lithuania |
| | 12 | Political Roadm | nap on l | mplementing the Synchronisation of | of the Baltic States' Electricity Networks with the Continental European |
| | | Network via Po | land | | |
| State governance level | | | | | |
| Ministry of Economics | | | 13 | National Energy and Climate Pla | a 2020 (NECP 2020) |
| | | | 14 | National Energy and Climate Plan | a 2030 (NECP 2030) |
| | | | | Annex I: Detailed information abo | out the connection of the plan with policy planning documents of Latvia |
| | | | | Annex II: Detailed information ab | out the consistency of the common result indicators of the ERDF and the |
| | | | | CF with the indicators of the Plan | l |
| | | | | Annex III: Information about the r | nain policies included in the baseline scenario of forecasts and their main |
| | | | | implementation measures | |
| | | | | Annex IV: Breakdown of policy n | easures |
| | | | | Annex V: Information on the nec | essary additional financing for the measures included in Annex 4 to the |
| | | | | plan with an identified amount o | f State budget financing |
| | | | | Annex VI: The methods used to a emissions | nalyse and forecast the development of energy systems and GHG |
| | | | 15 | Cabinet Order 04.02.2020 Nr. 46 | (about NECP 2021-2030) |
| | | | 16 | Energy Development Guidelines | 2016–2020 |
| | | | 17 | National Industrial Policy Guidel | nes 2021–2027 |
| Ministry of Education | and Sci | ience | 18 | Research and Innovation Strategy | for Smart Specialization (RIS3) |
| Ministry of Environme Regional Developr | ntal Pro | otection and | 19 | Strategy for Latvia for the Achiev | ement of Climate Neutrality by 2050 |
| Seeima of the Republic of Latvia | | 20 | Sustainable Development Strateg | v of Latvia until 2030 | |
| Cross-Sectoral Coordination centre | | 21 | National Development Plan of La | via 2021–2027 (NDP2027) | |
| EBRD | | | 22 | Strategy for Latvia | |
| Local governance level | l | | | | |
| Riga Energy Agency | | | | 23 | Riga Smart City Sustainable Energy Action Plan 2014–2020 |
| Riga Energy Agency | | | | 24 | Riga City Energy and Climate Plan 2022–2030 |
| Riga City Developmen | t Depar | rtment | | 25 | Sustainable Development Strategy Riga 2030 |

Table A2

Factiva search terms.

| Category | Criteria used |
|--------------------|---|
| Search terms | "energy" OR "renewable" OR "renewable energy" OR "wind" OR "solar" AND "electricity" AND "Latvia" |
| Search period | 01.06.2018–31.12.2021 |
| Sources | Latvian News Agency (to limit scope from over 3000 articles to 673) |
| Region | Latvia |
| Other restrictions | Headline and Lead paragraph |

Table A3

Illustrative interview guidelines/coding frame.

| VPM characteristics | VPM assessment criteria | Interview questions |
|-------------------------------|-------------------------|--|
| Policy element consistency | (a) | To what extent are the strategic goals for the transition aligned / not aligned between the EU and Latvia? To what extent is the energy transition aligned / not aligned with other strategic economic development goals in Latvia? To what extent is the energy transition strategic goal reflected in local level strategies? |

(continued on next page)

Table A3 (continued)

| VPM characteristics | VPM assessment criteria | Interview questions |
|--|----------------------------|---|
| | (b) | To what extent are the strategic goals substantiated with concrete policies (policy instruments) at the state and local levels? |
| | | To what extent is the state taking advantage of all support mechanisms available at the EU level, such as funding sources? |
| | | Can you identify any support mechanisms that would be helpful but currently do not exist at the EU level? |
| | (c) | Can you identify any contradictions between existing policies (policy instruments)? Consider, for instance, their target, duration, support level, stringency, differentiation. |
| | | Can you identify any policy instruments that are not used but would be helpful to advance the strategic goals? |
| Policy process coherence | (d) | How effective is policy planning between the EU and state levels? |
| | | How effective is policy planning between state and local levels? |
| | (e) | How effective is policy coordination between the EU and state levels? |
| | (A) | How effective is policy coordination between state and local levels? |
| | (1) | How effective is strategic and policy communication between the EU and state levels? Consider all policy |
| | | planning and implementation stages, including agenda setting, policy formulation, policy adoption. |
| | | How effective is strategic and policy communication between the state and local levels? Consider all policy |
| Policy mix | (g) | To what extent does the policy mix create a level playing field for renewable energy technologies in the |
| comprehensiveness | (g) | market? Why? |
| ·· · · · · · · · · · · · · · · · · · · | | To what extent does the policy mix address the relevant regulatory and administrative institutional issues? |
| | | Why? |
| | | To what extent does the policy mix enable the inclusion of renewable technologies in the existing energy system (e.g., infrastructure)? Why? |
| | (h) | To what extent does the policy mix support the supply side (technology-push) of renewable energy |
| | | technologies? Why? |
| | | To what extent does the policy mix support the demand side (technology-pull) for renewable energy |
| | | technologies? Why? |
| | (i) | How would you assess the diversity of policy instruments employed to advance the transition (economic, regulatory, soft)? Why? |
| Policy mix credibility | (i) | To what extent is the policy mix stable? Why? |
| | <u>.</u> | To what extent is the policy mix sufficiently flexible to enable change if required? Why? |
| | | To what extend does the policy mix create an attractive environment to investors and other business |
| | | stakeholders in renewable energy? Why? |

Table A4

Additional illustrative interview quotes.

| | 1 | |
|---|-----------------------|--|
| Assessment criteria | VPM frictior point | n Further relevant quotes |
| (I) Consistency of policy ele | ements | |
| Alignment of strategic priorities | SN—NA | "Latvia's plan goes in the right direction, identifies problems, but the ambition level does not exceed the EU recommendations" (EU2) |
| | NA-LO | "[Municipalities] would like to be more pro-actively involved in policy planning processes" (L2) |
| Instrument-strategy alignment | SN-NA | "Specificity and roadmaps for policies is lacking, also the contribution from the key economic sectors [] I am not certain that [the existing policy measures] will be sufficient at the end of the day" (EU2) |
| | NA-LO | "MPC has been artificial. Instead of creating a stimulus, it has been expensive" (N5) "We don't need any kind of public support. We just need a flat playing field where we can operate" (O2) |
| Inter-instrument alignment | SN—NA | "Public money does not attract private money. There is a lack of wise financial instruments. If Latvia continues to sleep on EU grant money and does not work with business angels, it will fall behind. That is an important factor – access to risk money." (EU2) |
| | NA-LO | "The private sector [] having issues with construction boards, that was the end of renewable energy in Latvia" (N6) |
| (II) Coherence of policy pro | ocesses | |
| Alignment of strategic SN—1 planning NA-Lu | | "Latvia positively and loudly supports 'Fit for 55' by EC and also has changed the usual caution in the rhetorics" (EU1) "Fossil gas is the largest enemy of a fair transition to clean energy in Latvia [] If there were no crippled market whereby gas is mysteriously cheaper, then renewable energy would long have spiked" (L1) |
| | | "Why could the largest company, that belongs to the state, not take on an ambitious renewable energy plan? Compared to Ignitis and Enefit Green, we are sitting around, waiting for something" (L1) |
| Alignment of coordination | SN—NA NA-LO | "EC controls whether Latvia lives up to the minimal requirements, and in general, these fit the general strategy" (EU2) "Need one institution to coordinate the green course [] municipalities are detached from the important topics, such as energy" (N5) |
| | | "We consult [the municipalities on relevant questions] as much as we are being asked" (N1) |
| Alignment of communication | SN—NA | "Industries have upheld the cliché that we are green, although that is not the case. We buy brown energy. If we are so green in the formal statistics – then why are we struggling so much?" (N6) |
| | | |

Table A4 (continued)

| (II) Coherence of policy processes | | | | | | |
|--|---|---|--|--|--|--|
| NA-LO | "The MPC has left the impression on citizens that renewable energy is the most expensive" (EU1) "There are many unused opportunities to make consumers more aware" (EU3) | | | | | |
| (III) Comprehensiveness | | | | | | |
| Market, institutional, and systemic failures addressed Technology-push and market-pull effects addressed Economic, regulatory, and soft instruments addressed | SN—NA- LO SN—NA- LO SN—NA | "Framework for innovation is currently in the planning process" (N3) "Innovation initiatives start with EU funds" (L3) "We see solar technologies as complementary to wind technologies, a fantastic solution in the context of decentralisation demands" (N1) "Tax policy changes are more important, more efficient, and stronger than CF support" (N4) | | | | |
| (IV) Credibility | | | | | | |
| Market, institutional, and systemic failures addressed | SN—NA- LO | "Structural factors come into play, for instance, governments have frequently changed, and decision- making in fragmented coalitions has been difficult [] Problem that cannot be solved – party relations in the government coalition" (EU1) "The energy ministers in Latvia have changed frequently, regulation in the field is not stable. We have made short term decisions [] The energy market situation is unattractive to foreign investors." (L1) "Problem exists because banks are cautious with project development. They do not see state politics. They see risks in how projects, such as wind parks, are implemented, as they remain stuck for ten years. If there were no other obstacles, finance would not be a problem, from bank side there is a big interest for good | | | | |

projects." (N6)

Table A5

Articles from Factiva search used for fact checking or referenced in text.

| Article title | Author | News | Date |
|--|--------------------|--------|--------------------------|
| | | agency | |
| There are parties in government coalition that are against touching VAT on energy - Gerhards | Silvija Reinberga | LETA | 20 December |
| There are parties in government countrol that are against touching viri on chergy occhards | biivija itemberga | | 2021 |
| Latvia lacks long-term strategy on energy - Wind Energy Association | n.a. | LETA | 15 December |
| Latita acto fong term blattegy on energy while Lifergy hobelation | | | 2021 |
| Latvenergo Group's turnover increases 17% in nine months | Marta Kronberga | LETA | 30 November |
| | 0 | | 2021 |
| Latvenergo planning to double capacity for generation of energy from renewable resources | Marta Kronberga | LETA | 29 November |
| | | | 2021 |
| Economics Ministry proposes cutting MPC in electricity bills by 65% in 2022 | n.a. | LETA | 5 November |
| | | | 2021 |
| Latvenergo sells by 70% more natural gas last year | Krišs Janis Kairis | LETA | 7 March 2021 |
| Audit Office to turn to authorities over EUR 0.5 million paid to two companies in mandatory | Raivis Spalvens | LETA | 7 January 2021 |
| procurement support | | | |
| Latvia has largely implemented recommendations in energy sector - European Commission | n.a. | LETA | 15 October 2021 |
| Toms Naburgs to be head of Latvian Wind Energy Association | n.a. | LETA | 15 September |
| | | | 2021 |
| Two Economics Ministry officials face prosecution for poor MPC oversight | Gatis Kristovskis | LETA | 18 May 2021 |
| Latvia has huge potential in wind energy - Levits | Baiba Krastina | LETA | 29 April 2021 |
| Elektrum solar energy park opened in Lithuania | Krišs Jānis Kairis | LETA | 7 April 2021 |
| Latvian-owned Elektrum Lithuania stepping into remote solar farm market | n.a. | LETA | 9 July 2020 |
| One third of Latvia's households ready to generate own power - Enefit | n.a. | LETA | 10 April 2019 |
| Karins assumes MPC might not be liquidated in March due to three important risks | n.a. | LETA | 20 March 2019 |
| Enefit to start selling electricity to Latvia's households in March | n.a. | LETA | 27 February |
| | | | 2019 |
| Ways to promote "green" energy should be found after scrapping MPC – economics minister | n.a. | LETA | 19 February |
| Latric has third largest share of renovables in more preven consumption. Furnetat | - | LETA | 2019 10 Februari |
| Latvia has third largest share of renewables in gross energy consumption - Eurostat | n.a. | LEIA | 12 February |
| Wind now inductor's development in Letvis homograd by lask of political vision essentiation | | IETA | 2019 20. January 2010 |
| Whild power industry's development in Latvia nampered by lack of pointical vision - association | II.a. | LEIA | 29 January 2019 |
| 85 2% of Latvia's residents support solar power development survey | II.d. | LEIA | 29 January 2019 |
| No need for new support mechanism to substitute MDC - economics minister nominee | n a | LEIA | 13 January 2019 |
| 27 "green" nower plants lose MDC licenses after snap inspections | n.a. | LETA | 3 January 2019 |
| 27 Site power plants lose will encenses after shap inspections Baltic Nordic countries sign agreement on cooperation in energy research program | n.a. | LETA | 1 October 2018 |
| Baltic, Nordic countries to increase energy-related research cooperation | n.a. | LETA | 28 September |
| parte, northe countries to increase energy-related research cooperation | 11.0. | | 2018 |
| Saeima committee in principle supports proposals on restricting MPC | na | LETA | 26 September |
| Suchina committee in principle supports propositio on reducting in G | | | 2018 |
| Nemiro suspends Economics Ministry's state secretary alleging violations in MPC oversight | | LETA | 5 February 2020 |
| ······································ | | | |

(continued on next page)

Table A5 (continued)

| Article title | Author | News | Date |
|---|-------------------------|--------|---------------------|
| | | agency | |
| | Raivis Spalvens, Ainars | | |
| | Leijejs | | |
| Climate and energy plan 2030 is step towards climate neutrality - ministers | Raivis Spalvens | LETA | 28 January 2020 |
| Government subsidies suspended for 21 green energy producers in 2019, saving EUR 31.5 million | Raivis Spalvens | LETA | 23 January 2020 |
| Work on removing MPC from electricity bills to be activated in the future - Nemiro | n.a. | LETA | 22 January 2020 |
| Production of wind energy should be based on market principles - Nem | Marta Kronberga | LETA | 3 December |
| | | | 2019 |
| Climate-neutral economy does not mean halt to economic growth - Puce | Raivis Spalvens | LETA | 21 November 2019 |
| Climate policy is opportunity to improve Latvia's welfare - Karins | Raivis Spalvens | LETA | 21 November |
| | | | 2019 |
| Latvia plans to ensure a share of at least 50% renewable energy in its overall final energy | Ivars Motivans | LETA | 8 November |
| Consumption of a second logic data 4 EV in 2019 | | LETA | 2019 E Contombor |
| Consumption of renewable energy sources drops 4.5% in 2018 | II.a. | LEIA | 2019 |
| Saeima investigative commission requests information on each individual MPC license and | n.a. | LETA | 16 August 2019 |
| officials' reasons for issuing licenses | | | |
| Government subsidies suspended for 134 green energy producers | n.a. | LETA | 1 July 2019 |
| "Green certificates" to be removed from MPC reform plan - PM Kucinskis | n.a. | LETA | 13 August 2018 |
| Government to make decision on Economics Ministry's proposal to scrap MPC in three years | n.a. | LETA | 13 August 2018 |
| PM Kucinskis to ask prosecutor's office to look into unlawful issue of MPC licenses | n.a. | LETA | 8 August 2018 |
| Nelja Energia: Latvian renewable subsidies have failed to develop sector | n.a. | LETA | 7 August 2018 |
| MPC liquidation would reduce electricity bills by 10% - Aseradens | n.a. | LETA | 2 August 2018 |
| Work group not to agree on liquidation of MPC system but work on renewing sector's reputation - expert | Raivis Spalvens | LETA | 27 July 2018 |
| Parliament orders PM to produce report on officials responsible for flawed green energy support | n.a. | LETA | 20 June 2018 |
| system Security of energy supply is and will remain priority of Latvia and EU - President Vejonis | n.a. | LETA | 1 June 2018 |

References

- Āboltiņš, R., 2019. A Breath of Fresh air: Analysis of Factors Affecting Deployment of Wind Energy in Latvia and Potential Solutions. CEE Bankwatch Network. https://bankwatch.org/wp-content/uploads/2019/12/breath-of-fresh-air.pdf.
- Āboltiņš, R., Akule, D., 2014. Gāzes Tirgus Liberalizācija Latvijā: Situācija Un Izaicinājumi. Providus. https://providus.lv/article_files/2754/original/gaze_FINAL-4. pdf?1418199364.
- Andonova, L.B., Tuta, I.A., 2014. Transnational networks and paths to EU environmental compliance: evidence from new member states. JCMS: J. Common Market Stud. 52 (4), 775–793. https://doi.org/10.1111/jcms.12126.
- Andonova, L.B., 2004. Transnational Politics of the Environment: The European Union and Environmental Policy in Central and Eastern Europe. MIT Press. Ashoff, G., 2005. Enhancing Policy Coherence For Development: Justification, Recognition and Approaches to Achievement. Deutsches Institut für

Entwicklungspolitik. https://www.econstor.eu/bitstream/10419/199136/1/die-study-11.pdf.

AST, 2022. Synchronisation With Europe. AS Augstsprieguma tikls. https://www.ast.lv/en/projects/synchronisation-europe.

Bach, H., Hansen, T., 2021. Flickering guiding light for maritime decarbonisation: past developments and future implications of the policy mix consistency and comprehensiveness for international shipping. In: Proceedings of the 12th International Sustainability Transitions Conference: Mainstreaming Sustainability Transitions: From Research Towards Impact. https://portal.research.lu.se/en/publications/flickering-guiding-light-for-maritime-decarbonisation-past-develo.

Bolwig, S., Bazbauers, G., Klitkou, A., Lund, P.D., Blumberga, A., Gravelsins, A., Blumberga, D., 2019. Review of modelling energy transitions pathways with application to energy system flexibility. Renew. Sustain. Energy Rev. 101, 440–452. https://doi.org/10.1016/j.rser.2018.11.019.

Borgström, S., 2019. Balancing diversity and connectivity in multi-level governance settings for urban transformative capacity. Ambio 48 (5), 463–477. https://doi. org/10.1007/s13280-018-01142-1.

Brauna, A., Sprance, I., Mače, Z., Ģelzis, Ģ., 2022. Kas Lobē «Gazprom» Intereses Latvijā? LSM. https://www.lsm.lv/raksts/zinas/latvija/kas-lobe-gazprom-intereses-latvija.a448170/.

Ćetković, S., Buzogány, A., 2019. The political economy of EU climate and energy policies in central and eastern Europe revisited: shifting coalitions and prospects for clean energy transitions. Politics Gov. 7 (1), 124–138. https://doi.org/10.17645/pag.v7i1.1786.

Cabinet of Ministers LV. (2019). Par Nacionalo energetikas Un Klimata Padomi. https://likumi.lv/ta/id/311155-par-nacionalo-energetikas-un-klimata-padomi.

Capano, G., Howlett, M., 2020. The knowns and unknowns of policy instrument analysis: policy tools and the current research agenda on policy mixes. Sage Open 10 (1), 215824401990056.

Carbon Market Watch. (2018). National energy and climate plans and the transition to carbon-free societies. https://carbonmarketwatch.org/wp/wp-content/uploads/2018/09/NATIONAL-ENERGY-AND-CLIMATE-PLANS-AND-THE-TRANSITION-TO-CARBON-FREE-SOCIETIES-_-WEB_final.pdf.

Chaisty, P., Whitefield, S., 2015. Attitudes towards the environment: are post-communist societies (still) different? Environ. Polit. 24 (4), 598–616. https://doi.org/ 10.1080/09644016.2015.1023575.

Cherp, A., Jewell, J., Goldthau, A., 2011. Governing global energy: systems, transitions, complexity. Glob. Policy 2 (1), 75–88. https://doi.org/10.1111/j.1758-5899.2010.00059.x.

Connoly, K., 2022. Germany to Reactivate Coal Power Plants As Russia curbs Gas Flow. The Guardian. https://www.theguardian.com/world/2022/jul/08/germany-reactivate-coal-power-plants-russia-curbs-gas-flow.

Costantini, V., Crespi, F., Palma, A., 2017. Characterizing the policy mix and its impact on eco-innovation: a patent analysis of energy-efficient technologies. Res Policy 46 (4), 799–819. https://doi.org/10.1016/j.respol.2017.02.004.

Cross-Sectoral Coordination Center. (2020). National development plan of Latvia for 2021-2027. https://pkc.gov.lv/sites/default/files/inline-files/NAP2027_ENG. pdf.

Dale, A., Robinson, J., King, L., Burch, S., Newell, R., Shaw, A., Jost, F., 2020. Meeting the climate change challenge: local government climate action in British Columbia, Canada. Clim. Policy 20 (7), 866–880. https://doi.org/10.1080/14693062.2019.1651244.

Del Río, P., 2007. The interaction between emissions trading and renewable electricity support schemes. An overview of the literature. Mitig. Adapt. Strateg. Glob. Change 12 (8), 1363–1390. https://doi.org/10.1007/s11027-006-9069-y.

- Del Río, P., 2014. On evaluating success in complex policy mixes: the case of renewable energy support schemes. Policy Sci. 47 (3), 267–287. https://doi.org/ 10.1007/s11077-013-9189-7.
- Dreblow, E., Duwe, M., Wawer, T., Donat, L., Zelljadt, E., Ayers, A., Upatniece, I., 2013. Assessment of Climate Change Policies in the Context of the European Semester: Country Report – Latvia. Ecologic Institute, Eclareon. https://www.ecologic.eu/sites/default/files/publication/2013/Country-report-Latvia-Assessment-of-climate-change-policies-in-the-context-of-the-European-semester_2013_en.pdf.

EBRD. (2016). Strategy for Latvia. https://www.ebrd.com/news/2016/ebrd-adopts-new-strategy-for-latvia.html.

- Edmondson, D.L., Kern, F., Rogge, K.S., 2019. The co-evolution of policy mixes and socio-technical systems: towards a conceptual framework of policy mix feedback in sustainability transitions. Res. Policy 48 (10), 103555. https://doi.org/10.1016/j.respol.2018.03.010.
- Edmondson, D.L., Rogge, K.S., Kern, F., 2020. Zero carbon homes in the UK? Analysing the co-evolution of policy mix and socio-technical system. Environ. Innov. Soc. Transit. 35, 135–161. https://doi.org/10.1016/j.eist.2020.02.005.
- Ehnert, F., Kern, F., Borgström, S., Gorissen, L., Maschmeyer, S., Egermann, M., 2018. Urban sustainability transitions in a context of multi-level governance: a comparison of four European states. Environ. Innov. Soc. Transit. 26, 101–116. https://doi.org/10.1016/j.eist.2017.05.002.
- Ekodoma. (2021). Rīgas Pilsētas Enerģētikas Un Klimata Rīcības Plāns. https://rea.riga.lv/upload/media/default/0001/01/

1c18bc893ee7c2dadc0a5499cd6eb25dd2ca0ea9.pdf.

- European Commission. (2019). A European green deal. https://ec.europa.eu/info/strategy/priorities-2019-2024/european-green-deal_en.
- European Commission. (2020). Commission staff working document: assessment of the final national energy and climate plan of Latvia. https://ec.europa.eu/energy/ sites/ener/files/documents/staff_working_document_assessment_necp_latvia.pdf.
- European Commission. (2021). Ceremony marks launch of the third estonia-Latvia electricity interconnector. https://ec.europa.eu/info/news/ceremony-markslaunch-third-estonia-latvia-electricity-interconnector-2021-aug-26 en.
- European Commission. (2021). National energy and climate plans. https://ec.europa.eu/info/energy-climate-change-environment/implementation-eu-countries/ energy-and-climate-governance-and-reporting/national-energy-and-climate-plans_en.
- European Commission. (2022). Energy communities. https://energy.ec.europa.eu/topics/markets-and-consumers/energy-communities_en.

European Commission. (2022). Energy union. https://energy.ec.europa.eu/topics/energy-strategy/energy-union_en.

- European Parliament. (2021). Revision of the renewable energy directive: fit for 55 package. https://www.europarl.europa.eu/thinktank/en/document/EPRS_BRI (2021)698781.
- Eurostat. (2020). Share of renewable energy in gross final energy consumption by sector. https://ec.europa.eu/eurostat/databrowser/view/SDG_07_40_custom_ 2191007/bookmark/table?lang=en&bookmarkId=61985b54-14e7-465a-9bfe-b7e7ceca9171.

Eurostat. (2020). SHARES tool. https://ec.europa.eu/eurostat/web/energy/data/shares.

- Eurostat. (2021). Share of energy from renewable resources. https://ec.europa.eu/eurostat/databrowser/view/NRG_IND_REN_custom_1216190/default/table? lang=en.
- Flanagan, K., Uyarra, E., Laranja, M., 2011. Reconceptualising the 'policy mix' for innovation. Res. Policy 40 (5), 702–713. https://doi.org/10.1016/j. respol.2011.02.005.
- George, A.L., Bennett, A., 2005. Case Studies and Theory Development in the Social Sciences. MIT Press.
- Grigas, A., 2013a. Energy Policy: the Achilles Heel of the Baltic States. Jacques Delors Institute. https://institutdelors.eu/wp-content/uploads/2018/01/ balticstateseu-energypolicy-grigas-ne-jdi-july13.pdf.
- Grigas, A., 2013b. The Politics of Energy and Memory Between the Baltic states and Russia. Post-Soviet Politics. Routledge.
- Grubb, M., McDowall, W., Drummond, P., 2017. On order and complexity in innovations systems: conceptual frameworks for policy mixes in sustainability transitions. Energy Res. Soc. Sci. 33, 21–34. https://doi.org/10.1016/j.erss.2017.09.016.
- Hale, T., 2020. Transnational actors and transnational governance in global environmental politics. Ann. Rev. Political Sci. 23 (1), 203–220. https://doi.org/10.1146/ annurev-polisci-050718-032644.
- Howlett, M., Del Río, P., 2015. The parameters of policy portfolios: verticality and horizontality in design spaces and their consequences for policy mix formulation. Environ. Plann. C Govern. Policy 33 (5), 1233–1245. https://doi.org/10.1177/0263774X15610059.
- Howlett, M., Rayner, J., 2007. Design principles for policy mixes: cohesion and coherence in 'New governance arrangements'. Policy Soc. 26 (4), 1–18. https://doi.org/10.1016/S1449-4035(07)70118-2.
- Howlett, M., Vince, J., Del Río, P., 2017. Policy integration and multi-level governance: dealing with the vertical dimension of policy mix designs. Politics Govern. 5 (2), 69–78. https://doi.org/10.17645/pag.v5i2.928.
- Huang, P., 2019. The verticality of policy mixes for sustainability transitions: a case study of solar water heating in China. Res. Policy 48 (10), 103758. https://doi.org/10.1016/j.respol.2019.02.009.
- IEA, 2022. What is Behind Soaring Energy Prices and What Happens next? March 16 International Energy Agency. https://www.iea.org/commentaries/what-is-behind-soaring-energy-prices-and-what-happens-next.
- Irbe, J. (2022). Latvenergo AS will invest EUR 200 million in sustainable energy projects. https://view.news.eu.nasdaq.com/view? id=bfaa08106df515f7e20566d92c798bbda&lang=en.
- Kamenders, A., Rochas, C., Juergens, I., Rusnok, D., 2020. Nepieciešamās Investīcijas Latvijas enerģētikas Un Klimata Mērķu 2030 Izpildei. Riga Technical University. https://videszinatne.rtu.lv/wp-content/uploads/2020/05/Nepiecie%C5%A1am%C4%81s-invest%C4%ABcijas-Latvijas-ener%C4%A3%C4%93tikas-un-klimatam%C4%93r%C4%B7u-2030-izpildei.pdf.
- Kern, F., Rogge, K.S., Howlett, M., 2019. Policy mixes for sustainability transitions: new approaches and insights through bridging innovation and policy studies. Res. Policy 48 (10), 103832. https://doi.org/10.1016/j.respol.2019.103832.
- Kincis, J., 2023. Latvia's Climate and Energy Ministry Begins Work. March 1. LSM. https://eng.lsm.lv/article/politics/politics/latvias-climate-and-energy-ministrybegins-work.a489930/.
- Kingdon, J.W., Stano, E., 2011. Agendas, alternatives, and public policies, 2nd edition. Longman.
- Kivimaa, P., Kern, F., 2016. Creative destruction or mere niche support? Innovation policy mixes for sustainability transitions. Res. Policy 45 (1), 205–217. https://doi.org/10.1016/j.respol.2015.09.008.
- Kivimaa, P., Sivonen, M.H., 2021. Interplay between low-carbon energy transitions and national security: an analysis of policy integration and coherence in Estonia, Finland and Scotland. Energy Res. Soc. Sci. 75, 102024 https://doi.org/10.1016/j.erss.2021.102024.
- Kivimaa, P., Virkamäki, V., 2014. Policy mixes, policy interplay and low carbon transitions: the case of passenger transport in Finland. Environ. Policy Govern. 24 (1), 28–41. https://doi.org/10.1002/eet.1629.
- Klūga, M., 2022. Latvijā Aizliedz Dabasgāzes Piegādes No Krievijas. July 14. LSM. https://www.lsm.lv/raksts/zinas/ekonomika/latvija-aizliedz-dabasgazes-piegadesno-krievijas.a465404/.
- Knill, C., Liefferink, D., 2021. The establishment of EU environmental policy. Eds. In: Jordan, A., Gravey, V. (Eds.), Environmental Policy in the EU: Actors, Institutions and Processes. Routledge, pp. 13–32. https://doi.org/10.4324/9780429402333-3.
- Kuzemko, C., Lockwood, M., Mitchell, C., Hoggett, R., 2016. Governing for sustainable energy system change: politics, contexts and contingency. Energy Res. Soc. Sci. 12, 96–105. https://doi.org/10.1016/j.erss.2015.12.022.
- Kuzemko, C., Bradshaw, M., Bridge, G., Goldthau, A., Jewell, J., Overland, I., Scholten, D., van de Graaf, T., Westphal, K., 2020. Covid-19 and the politics of sustainable energy transitions. Energy Res. Soc. Sci. 68, 101685 https://doi.org/10.1016/j.erss.2020.101685.
- Lanahan, L., Feldman, M.P., 2015. Multilevel innovation policy mix: a closer look at state policies that augment the federal SBIR program. Res. Policy 44 (7), 1387–1402. https://doi.org/10.1016/j.respol.2015.04.002.
- Latvian Wind Energy Association, 2022. Latvia Has a Lot of Untapped Potential in the Wind Sector. August 6. Baltic Wind. https://balticwind.eu/latvia-has-a-lot-ofuntapped-potential-in-the-wind-sector/.

Lindberg, M.B., 2019. The EU emissions trading system and renewable energy policies: friends or foes in the European policy mix? Politics Govern. 7 (1), 105–123. https://doi.org/10.17645/pag.v7i1.1800.

Lindberg, M.B., Markard, J., Andersen, A.D., 2019. Policies, actors and sustainability transition pathways: a study of the EU's energy policy mix. Res. Policy 48 (10), 103668. https://doi.org/10.1016/j.respol.2018.09.003.

Lindroos, T.J., Lehtilä, A., Koljonen, T., Kofoed-Wiuff, A., Hethey, J., Dupont, N., Vitina, A., 2018. Baltic Energy Technology Scenarios. Nordic Council of Ministers. https://doi.org/10.6027/TN2018-515.

Loorbach, D., 2010. Transition management for sustainable development: a prescriptive, complexity-based governance framework. Governance 23 (1), 161–183. https://doi.org/10.1111/j.1468-0491.2009.01471.x.

Magro, E., Wilson, J.R., 2013. Complex innovation policy systems: towards an evaluation mix. Res. Policy 42 (9), 1647–1656. https://doi.org/10.1016/j.

Magro, E., Wilson, J.R., 2019. Policy-mix evaluation: governance challenges from new place-based innovation policies. Res. Policy 48 (10), 103612. https://doi.org/ 10.1016/j.respol.2018.06.010.

Markard, J., Rosenbloom, D., 2020. Political conflict and climate policy: the European emissions trading system as a Trojan Horse for the low-carbon transition? Clim. Policy 20 (9), 1092–1111. https://doi.org/10.1080/14693062.2020.1763901.

Markard, J., Raven, R., Truffer, B., 2012. Sustainability transitions: an emerging field of research and its prospects. Res. Policy 41 (6), 955–967. https://doi.org/ 10.1016/j.respol.2012.02.013.

Meadowcroft, J., 2011. Engaging with the politics of sustainability transitions. Environ. Innov. Soc. Transit. 1 (1), 70–75. https://doi.org/10.1016/j.eist.2011.02.003.
Meckling, J., Sterner, T., Wagner, G., 2017. Policy sequencing toward decarbonization. Nat. Energy 2 (12), 918–922. https://doi.org/10.1038/s41560-017-0025-8.
Ministry of Economics LV. (2018). EC reconfirms that support to manufacturers of RE in Latvia is legal. https://www.google.com/search?q=EC+reconfirms+that+support+to+manufacturers+of+RE+in+Latvia+is+legal&rlz=1C1GCEA_enCH820CH820&oq=EC+reconfirms+that+support+to+manufacturers+of+RE+in+

Latvia+is+legal&aqs=chrome.69i57j69i64.285j0j4&sourceid=chrome&ie=UTF-8.

Ministry of Economics LV. (2020). National energy and climate plan 2021-2030. https://ec.europa.eu/energy/sites/ener/files/documents/lv_final_necp_main_en.pdf. Ministry of Education and Science LV. (2018). Monitoring of smart specialization strategy: informative report. https://www.izm.gov.lv/en/media/3745/download. Morgan, S. (2018). EU, Baltics, Poland target Russia grid separation By 2025. Euractiv. https://www.euractiv.com/section/electricity/news/eu-baltics-poland-targetrussia-grid-separation-by-2025/.

Nauwelaers, C., Boekholk, P., Mostert, B., Cunningham, P., Guy, K., Hofer, R., Rammer, C., 2009. Policy Mixes For R&D in Europe. University of Manchester Research. https://www.research.manchester.ac.uk/portal/files/33081654/FULL TEXT.PDF.

Newbold, C., Boyd-Barrett, O., van Bulck, H.den (Eds.), 2002. The Media Book. Arnold.

Newell, S.J., Goldsmith, R.E., 2001. The development of a scale to measure perceived corporate credibility. J. Bus. Res. 52 (3), 235–247. https://doi.org/10.1016/ S0148-2963(99)00104-6.

Nilsson, M., Zamparutti, T., Petersen, J.E., Nykvist, B., Rudberg, P., McGuinn, J., 2012. Understanding policy coherence: analytical framework and examples of sector-environment policy interactions in the EU. Environ. Policy Gov. 22 (6), 395–423.

Oberlander, J., Weaver, R.K., 2015. Unraveling from within? The affordable care act and self-undermining policy feedbacks. Forum (1), 37–62. https://www.degruyter.com/document/doi/10.1515/for-2015-0010/html.

OECD, 2019a. Environmental Performance Reviews: Latvia. Organization for Economic Cooperation and Development. https://doi.org/10.1787/2cb03cdd-en.

OECD, 2019b. Recommendation of the Council on OECD Legal Instruments Policy Coherence For Sustainable Development. Organization for Economic Cooperation and Development. https://www.oecd.org/gov/pcsd/recommendation-on-policy-coherence-for-sustainable-development-eng.pdf.

Ossenbrink, J., Finnsson, S., Bening, C.R., Hoffmann, V.H., 2019. Delineating policy mixes: contrasting top-down and bottom-up approaches to the case of energystorage policy in California. Res. Policy 48 (10), 103582. https://doi.org/10.1016/j.respol.2018.04.014.

Oxford Institute for Energy Studies. (2022, March 8). Ukraine invasion: what this means for the European gas market. https://www.oxfordenergy.org/publications/ ukraine-invasion-what-this-means-for-the-european-gas-markets/.

Reichardt, K., Negro, S.O., Rogge, K.S., Hekkert, M.P., 2016. Analyzing interdependencies between policy mixes and technological innovation systems: the case of offshore wind in Germany. Technol. Forecast. Soc. Change 106, 11–21. https://doi.org/10.1016/j.techfore.2016.01.029.

Rogge, K.S., Dütschke, E., 2018. What makes them believe in the low-carbon energy transition? Exploring corporate perceptions of the credibility of climate policy mixes. Environ. Sci. Policy 87, 74–84. https://doi.org/10.1016/j.envsci.2018.05.009.

Rogge, K.S., Reichardt, K., 2016. Policy mixes for sustainability transitions: an extended concept and framework for analysis. Res. Policy 45 (8), 1620–1635. https://doi.org/10.1016/j.respol.2016.04.004.

Rogge, K.S., Kern, F., Howlett, M., 2017. Conceptual and empirical advances in analysing policy mixes for energy transitions. Energy Res. Soc. Sci. 33, 1–10. https://doi.org/10.1016/j.erss.2017.09.025.

Schimmelfennig, F., Sedelmeier, U. (Eds.), 2005. Cornell Studies in Political economy. The Europeanization of Central and Eastern Europe. Cornell University Press. Solorio Sandoval, I., Jörgens, H. (Eds.), 2017. A Guide to EU Renewable Energy policy: Comparing Europeanization and Domestic Policy Change in EU Member States. Edward Elgar Publishing.

Sovacool, B.K., 2009. The importance of comprehensiveness in renewable electricity and energy-efficiency policy. Energy Policy 37 (4), 1529–1541. https://doi.org/ 10.1016/j.enpol.2008.12.016.

Sprance, I., & Klava, B. (2019, June 26). OIK Cilpā. Ir. https://ir.lv/2019/06/26/oik-cilpa/.

Sprance, I. (2020, December 9). Sadod 'pa nagiem'. Ir. https://ir.lv/2020/12/09/sadod-pa-nagiem/.

SPRK. (2016). Kas Ir OIK Un Kāpēc Tā Ir Mūsu rēķinos? Sabiedrisko pakalpojumu regulēšanas komisija. https://lvportals.lv/skaidrojumi/281264-kas-ir-oik-un-kapec-ta-ir-musu-rekinos-2016.

Stemler, S., 2000. An overview of content analysis. Pract. Assess. Res. Eval. 7 (1) https://doi.org/10.7275/z6fm-2e34.

Strambo, C., Nilsson, M., Månsson, A., 2015. Coherent or inconsistent? Assessing energy security and climate policy interaction within the European Union. Energy Res. Soc. Sci. 8, 1–12. https://doi.org/10.1016/j.erss.2015.04.004.

Tanaka, Y., Chapman, A., Tezuka, T., Sakurai, S., 2020. Putting the process into the policy mix: simulating policy design for energy and electricity transitions in Japan. Energy Res. Soc. Sci. 70, 101702 https://doi.org/10.1016/j.erss.2020.101702.

Wright, D. (Ed.), 2008. Thinking in Systems: A Primer. Chelsea Green Publishing.

Yin, R.K., 2009. Case Study Research: Design and Methods, 5th edition. Sage Publications.

Zepa, I., 2022. From energy islands to energy highlands? Political barriers to sustainability transitions in the Baltic region. Energy Res. Soc. Sci. 93, 102809 https:// doi.org/10.1016/j.erss.2022.102809.